Author's response to reviews

Title: Interaction between polymorphisms of the Human Leukocyte Antigen and HPV-16 Variants on the risk of invasive cervical cancer

Authors:

Patricia S de Araujo Souza (psas@inca.gov.br)
Paulo C Maciag (maciag@advaxis.com)
Karina B Ribeiro (k bribeiro@gmail.com)
Maria Luiza Petzl-Erler (perler@ufpr.br)
Eduardo L Franco (eduardo.franco@mcgill.ca)
Luisa L Villa (llvilla@ludwig.org.br)

Version: 3  Date: 30 July 2008

Author's response to reviews: see over
July 30th, 2008

The Editor
BMC Cancer

Ref: MS submission

Dear Madam or Sir,

I enclose herewith a revised version of an original manuscript entitled “Interaction between polymorphisms of the Human Leukocyte Antigen and HPV-16 variants on the risk of invasive cervical cancer” by myself, Maciag, Ribeiro, Petzl-Erler, Franco and Villa.
Regarding the comments and questions addressed by the referees, I provide a list of answers below.

Sincerely yours,

Patrícia Savio de Araujo Souza, PhD
Responses to Dr Zehbe’s comments:

1. As suggested, we revised the types in manuscript.
Responses to Dr. Xu’s comments:

1. As suggested, we included tables 3 and 4 in the main text.